

Thesis title

EXPLORING FACULTY PERCEPTIONS OF THE VALUE OF ADVISORY BOARDS IN ACCOUNTING EDUCATION CURRICULUM DEVELOPMENT

Kgobalale Nebbel Motubatse Haaga-Helia University of Applied Sciences Master of Business Administration (MBA) Education Management Master's thesis 2024

Author(s) Kgobalale Nebbel Motubatse

Degree

Master of Business Administration: Education Management Report/Thesis title

Exploring faculty perceptions of the value of advisory boards in accounting education curriculum development

Number of pages + number of appendices 63 + 3

This study explores faculty members' perceptions of the value of advisory boards in accounting education curriculum development. Thus, the study explores the role of advisory boards in accounting curriculum development; the composition of advisory boards; and key barriers encountered by curriculum developers and advisory boards. This study is encouraged by the fact that there are inconsistencies in the appointment of advisory committees and in the roles they play in curriculum development, even though the University of Technology (UoT) has issued a clear policy directive on the appointment of academic advisory committees to ensure that the learning programs they support are of outstanding quality, professionally relevant, and aligned with the latest developments and technologies in use in the professions, businesses, and industries graduates will enter. The study follows a qualitative research method to gain information about the perceptions of HoDs, advisory board members, academic managers, and alumni/graduates of the role of the advisory board on maintaining curriculum relevance in UoT's Faculty of Economics and Finance. Semistructured interviews were used to collect data from the participants. The study was limited to respondents associated with the UoT, and the results could be used to improve the practical effectiveness of advisory boards in the university, and to assess (and improve) the current policy on advisory boards based on the perceptions of the research participants. The findings could enhance the role and use of advisory boards in the Faculty of Economics and Finance at UoT.

Keywords: Advisory boards, curriculum, accounting education, university, graduate attributes

Abbreviation

UoT: University of Technology HH: Haaga-Helia HoD: Head of Department HEDS-RICH-REC: Higher Education Development and Support, Research and Innovation Committee, Research Ethics Committee

TABLE OF CONTENTS

CHAP	PTER 1	6
INTRO	ODUCTION TO THE STUDY	6
1.1	Introduction	6
1.2	Research Problem	9
1.3	Research Questions	12
1.4	Aim of the Study	12
1.4.1	Objectives of the study	12
1.4.2	Rationale for the study	13
1.5	Outline of Thesis	13
CHAP	PTER 2	15
LITER	RATURE REVIEW	15
2.1	Introduction	15
2.2	Role of Advisory Boards	15
2.3	Composition of Advisory Boards	16
2.4	Key barriers faced by curriculum developers and advisory boards	17
2.5	Summary of the Chapter	18
CHAP	PTER 3	19
RESE	ARCH DESIGN AND METHODOLOGY	19
3.1	Introduction	19
3.2	Research Paradigm	19
3.3	Research Methodology	19
3.4	Case Study Research	20
3.5	Population of the study	21
3.6	Data collection technique	22
3.6.1	e-Interviews	23
3.6.2	Primary research questions	24
3.7	Data analysis technique	24
3.7.1	Data Coding	25
3.7.2	Coding and analysis	25
3.8	Trustworthiness	26
3.9	Ethical consideration	26
3.10	Summary of the Chapter	27

CHAP	TER 4	28
PRES	ENTATION OF RESULTS	28
4.1	Introduction	28
4.2	Section A: Profile of the Participants	28
4.3	Thematic Analysis	29
4.4	Summary of the chapter	39
CHAP	TER 5	41
CONC	LUSION AND RECOMMENDATIONS	41
5.1	Introduction	41
5.2	Overview of the study	41
5.3	Conclusions drawn from the data analysis	42
5.4	Implication of the study	45
5.5	Limitations of the study	46
5.6	Recommendations and future research	46
5.7	Final conclusion of the study	47
REFE	RENCES	48
	XURE 1	64

TABLES:

Table 3.1: Population and sampling	21
Table 4.1: Demographic information of the participants	29
Table 4.2: Themes and categories corresponding to interview questions	30

CHAPTER 1

INTRODUCTION TO THE STUDY

1.1 Introduction

In the last decade the University of Technology (UoT) has adopted a policy on advisory boards to facilitate transformation of the curriculum and to provide expert advice on teaching the transformed curriculum, and to assess its impact on students' learning experiences. Schwartz and Fogg (1985) define the advisory board/committee as a group of individuals who are appointed to provide advice and guidance to the academic management team on specific issues affecting the curriculum. The board/committee is generally made up of individuals from different areas of expertise who can offer a range of perspectives and insights on the issues faced by the departments. Advisory boards appointed by the UoT's Faculty of Economics and Finance have achieved significant compliance with the university's policy, supporting the need to maintain curriculum relevance, and implement work integrated learning, improve teaching and learning quality, and to contribute to improving the programs' standards so that accreditation with professional bodies is achieved/maintained. Taken together, advisory boards have had to provide support in a variety of areas where their specific skill set of leadership (Kearns, 2019) is able to assist the faculty to reposition, reinvent and rescale its teaching plans and strategies (Kilcrease, 2011; Mello, 2019). For context, the academic programs and teaching efforts have frequently been affected by unusual internal events (including violent student protests on campus), and external events (Beetham & Sharpe, 2019). Thus, these events required that the advisory boards advise and support the faculty in its efforts to deliver quality education (Kilcrease, 2011; Mello, 2019) in spite of the turmoil. Advisory boards support the education programme (McPeak, 2012; Gabbin et al., 2020; Albring & Elder, 2020; Berikol & Killi, 2021) through the identification of research opportunities (Kilcrease, 2011) and the maintenance of a community engagement programme (Boadi & Osarfo, 2019). Ultimately, the intention behind having advisory boards is to further advance academic standards and to support the quality of the programs (McPeak, 2012; Gabbin et al., 2020; Albring & Elder, 2020; Berikol & Killi, 2021).

The value of the role played by the advisory boards has often been underestimated, and critics have also contrasted boards' involvement with that of private individuals and public, professional

bodies in the management of teaching and curriculum development (Kilcrease, 2011; Zahra et al., 2011). According to Kilcrease (2011), advisory boards can provide new and effective insights into how academic programs can be brought about. The fact is that advisory boards influence the subjects' vocabulary (ensuring academia keeps its terminology accessible to professional practitioners), and this has led to the development of useful teaching terminologies. Many advisory board propositions do not really make changes the university policies, but through simplification, they enable multiple teaching and learning methodologies to be unified into a single useful process. This study then, explores the value of advisory boards specifically in accounting education at the UoT.

The literature on advisory boards in accounting education makes use of a variety of overlapping/interchangeable concepts (for example, see Gabbin, 2002; Kilcrease, 2011; McPeak, 2012; Gabbin et al., 2020; Albring & Elder, 2020; Berikol & Killi, 2021). And 'advisory boards' has become one of the most popular catchwords and research topics in the academic space. As with any term shared by a variety of authors from multiple academic disciplines, as well as being part of everyday language, 'advisory boards', as a term, is a concept overloaded by multiple definitions and contexts, trying to prescribe how the term should be used. The main thrust of the literature's research on advisory boards has been that they are a source of advice and support that is directed at building teaching and learning capacity. Yet, in researching this topic, the current study has found very little literature that specifically explores the value of advisory boards in accounting education and little in the way of explicit definitions of 'advice' and 'support' in the description. However, it can be safely said, with respect to advisory boards, that the 'advice' and 'support' they offer should be in the explanation of and motivation for curriculum development.

Curriculum debates have arisen in the major institutions as a response to aspects of discussions on the impact of societal development on academia (Yastrebov, 2016). Such debates have focused primarily on the form of professional advancement (Reid, 2013); career development (Healy et al., 2020); societal influence (Wrigley, 2018); technological change (Erstad and Voogt, 2018); field of knowledge (O'Connor, 2022); and commitment to specialized education (Sant, 2019). All these issues are still parts of a problem commonly connected with curriculum development, and particularly in the universities of technology. However, Willcocks (2017) states that this is not justified, because the curriculum should be used as a vehicle to empower students and graduates with the necessary proficiency, attitudes, and values to aid them in personal and career advancement. Thus, it should be seen as mandatory (not debatable) that a curriculum meets societal expectations for sustainable development. It is generally accepted that advisory boards in the field of accountancy remain the 'movers and shakers' of accountancy curriculum development, always motivating for greater change/responsiveness to society's needs. Arguably, the movers and shakers of curriculum development are not always inside the classroom but are nevertheless still connected to an educational institution.

Historically, the role of advisory boards has frequently been ignored in curriculum debates and less formal discussions. However, times have changed, and the impact of external voices is now appreciated and recognised as a pertinent source of members for advisory boards and panels. The universities of technology now recognize that, to prepare their students and graduates for easier assimilation into the job market, and for their being better able to demonstrate expected graduate attributes, the advisory roles of the professional bodies are more important than the universities' internal policies on curriculum development. Thus, advisory bodies are beginning to realize their potential, and are paying an important role in curriculum design, standards setting, establishing networks for professionals to meet and discuss their field of expertise with faculty members and students, and to develop/issue a code of conduct to guide professional behaviors.

Through this involvement graduate attributes, insights and skills are extended, and the labor market is more likely to pay attention to/employ graduates from such universities of technology. In today's world of employment, employers are interested in recruiting graduates who can show a higher work ethic, and already possess a greater repertoire of skills than the 'normally' academically qualified graduates. Although Harms (1974) describes teachers as change agents in curriculum design and implementation, professional bodies are responsible for curriculum management and that significantly affects how the qualification program impacts the labor market with respect to the desired attributes of their graduates.

One of the simplest and most effective ways of getting advisory boards to participate in curriculum development is simply to take every opportunity to talk. Whether the onus is on curriculum designers in the field to initiate the conversations, or that the responsibility rests with board members is not particularly relevant here: it remains an obvious starting point to get excellent advice from the advisory board. Supporting this idea, Kilcrease (2011) conducted a study on faculty perceptions of business advisory boards, and the challenges of making use of their services. Kilcrease surveyed 1600 business faculty members about their experiences of advisory boards. The findings reveal that faculties are not directly participating in the activities of advisory

boards, and some do not even attend advisory board meetings, not even to simply listen to the board members' contributions.

1.2 Research Problem

Despite there being a clear policy directive at UoT on the need to appoint academic advisory committees to ensure that learning programs are of outstanding quality, relevant, and aligned to the latest developments and technologies in the profession, business, and/or industry (TUT, 2008; and TUT, 2022), there are still some inconsistencies in the appointment of advisory committees and in defining the role to be played in curriculum development. Scholars such as Kilcrease (2011), Zahra et al., (2011), Boadi and Osarfo (2019), García-Meca (2019), and Abeysekera (2015) continue to argue that the role of advisory boards in curriculum development remains limited. That said, there are substantial gaps in the current knowledge base about how academic advisory boards function (Milman, 2017). One such gap is as a result of the board members' low participation in curriculum development (Van Hoof et al., 2017). In theoretical terms, advisory committees exist as a symbolic structure to ensure that learning programs are relevant to future workplaces. Although the existing policy on advisory boards provides key functional activities, with significant guidance in guiding curriculum changes and program development, the current study sought to understand the perceptions of the HoDs, subject coordinators/lecturers, advisory board members, student house committees, curriculum designers and alumni/graduates about how well advisory boards complete their tasks. As all the faculty's departments have now appointed advisory boards, the key research question is this: How do departments view advisory boards' usefulness in [ensuring the relevance of] the accounting education curriculum?

This study's contribution is that it builds on the more recently published studies in the field of accounting education in South African universities. Although a few studies have examined the role of advisory boards in accounting education (see, Gabbin, 2002; Kilcrease, 2011; McPeak, 2012; Gabbin et al., 2020; Albring & Elder, 2020; Berikol & Killi, 2021), the focus on the value of advisory boards in accounting education curriculum development in South African universities has been muted at best. Thus, this study provides additional insight into the types of input which could be contributed by the advisory boards. Advisory boards have been constituted so as to be able to tap into private individuals' and senior public sector officials' current knowledge of the discipline, and thus to provide direction on curriculum development, and by bringing with them new terminologies to explain new ways of teaching and learning. This research differs from

previous studies in that it starts by identifying the gaps in the literature and the current practices which affect expectations of the role of advisory boards within the Faculty of Economics and Finance at the UoT. In addition, the research identifies how the current practice deviates from the university's policy on advisory boards. In doing so, the study draws strongly on the work of Kilcrease (2011), who quantitatively surveyed over 1,600 business studies' advisory boards and found that faculty members did not attend advisory board meetings, nor did they value the contributions and potential importance of advisory boards for their faculties.

There are very few published studies that have attempted to illustrate the role of advisory boards in the development of the accounting curriculum. Those few recent studies have used quantitative, qualitative, and mixed methodologies in the study of academic advisory boards. These studies reveal that advisory boards primarily address risk management and curriculum content (Query, 2018), and that advisory boards have been able to improve accreditation standards and quality management (Richards et al., 2019).

The research shows advisory boards are usually asked to advise on curriculum content, to ensure that accounting graduates possess the skills needed by the market (Kilcrease, 2011; Zahra et al., 2011; Dewua & Barghathť, 2019; Snead et al., 2019; Gabbin et al., 2020; Albring & Elder, 2020; Berikol & Killi, 2021), and that advisory boards should therefore interact with curriculum committees and participate in faculty meetings (Norman & Bagranoff, 2019; Kilcrease, 2011). Thus, while these studies provide useful information on the key attributes desired of advisory boards, there may be other tasks and types of input that advisory boards could provide to positively influence curriculum development in the faculty.

Constructing a viable and pertinent curriculum is not a straightforward activity. It needs a broader perspective that can best be provided by involving a variety of stakeholders. According to Albrecht et al., (1994), university accounting programs have failed to recognize the extensive changes taking place in the outside world, largely brought about by new technology, changing societal values and government policies, and advancements in business practices and structures. As a result, current curricula do not address numerous skills and competencies that industry now believes should be taught in university accounting programs. Bonk and Smith (1998) identified the need for accounting departments to include higher order thinking skills in their curricula. A higher order thinking skill is referred to as something that is subject to lifelong learning, in that

new knowledge and skills necessary for successful adaptation to a changing world are continuously acquired throughout life (Forster, 2004).

Accountancy departments in some of the universities of technology in South Africa have established an effective relationship with their advisory boards. However, others are still struggling with the establishment of workable relationships (Imperial, 2005); the issue of the relatively low skill sets of members of advisory boards (Feld, 2002), and the challenge of attracting knowledgeable board members who are willing to give/donate their time (Yee, 2020) and knowledge to education (Rebele and Pierre, 2019), due to their 'day job' work commitments.

Tapis and Delaney (2017) remark that there is an existing gap between what academics deem necessary and appropriate to teach, and what skills practitioners require of their newly graduated employees: thus, the gap appears to be between the development of the curriculum (and the teaching of that curriculum) and what practitioners need/expect from recently graduated employees. This gap means that those who develop the curriculum are generally separate from those who employ the people who have been trained/educated according to the curriculum. According to Doyle (1992), the gap is seen as a separation of curriculum and teaching on one hand, from the required skillset of newly employed graduates. Graves (2021) warns curriculum developers to "mind the gap". Justifying this view, Tapis and Delaney (2017) observed that practitioners are dismissive of the current academic approach to curriculum development, and requested that advisory boards bridge the gap by advising and supporting the development of curriculum.

By inviting professional bodies (as representatives of industry) to participate in curriculum development ensures that graduates have abilities that fit the market demands (Trung, 2016). The recent relatively uncontrolled implementation of curriculum changes, without the input or approval of professional bodies, almost inevitably has had a negative impact on graduates' suitability for the job market. These professional bodies are therefore interested in, and essential for shaping key developments of the curriculum to meet industry's needs. Thus, for a curriculum to be successfully developed and implemented it needs professionals to oversee its compatibility with market needs.

1.3 Research Questions

In the light of the above discussion, this study aims to contribute to understanding the role of advisory boards by exploring faculty perceptions of the value of advisory boards in accounting education curriculum development.

The research questions were organized to explore three main areas:

- *RQ 1:* How do participants perceive the role of advisory boards in accounting curriculum development?
- RQ 2: What is the composition of advisory boards in the faculty?
- *RQ 3:* What are the challenges of the curriculum development in the faculty?

The responses to the above research questions were collected through electronic interviews (einterviews) with the participants, the overall objective being to explore their perceptions of the value of advisory boards in accounting education curriculum development.

1.4 Aim of the Study

This study aims to explore faculty perceptions of the value of advisory boards in the development of the accounting education curriculum.

1.4.1 Objectives of the study

The specific research objectives to support the main aim of the study are:

- to determine the role of advisory boards in accounting curriculum development.
- to establish/determine the composition of advisory boards.
- to identify key barriers faced by curriculum developers and advisory boards.

1.4.2 Rationale for the study

This qualitative case study is important and needed for several reasons. Firstly, there is an existing gap in the literature: the perceptions HoDs, academic managers, lecturers, advisory board members, and alumni/graduates are not coherent in that there is not a uniform understanding of the value of advisory boards in curriculum development. The available literature has only focused on the role of advisory boards in curriculum development by studying the composition of the boards. In the main, such literature has been mainly focused on quantitative research. The current study therefore aims to explore the other side of the process, the faculty members' perceptions of the value of advisory boards in the accounting education curriculum development process.

Secondly, the study has made use of focused qualitative findings to gain an in-depth understanding of the faculty perceptions of how the advisory boards are used to improve the accounting curriculum. This study was conducted at the University of Technology, and focused on those programs falling under the Faculty of Economics and Finance at the UoT. The faculty uses the UoT's approved policy on the formation of academic advisory committees; however, the implementation of the policy throughout the faculty is not uniform regarding composition/membership of the committees, and neither are the roles of the advisory committees equivalent.

Lastly, the study provides base-line information for future researchers who wish to study the developing roles of academic advisory boards in accounting education and curriculum development.

1.5 Outline of Thesis

This thesis is divided into five chapters.

Chapter 1 provides a brief introduction to the study, presents the research problem, research objectives, and the rationale for the study, and finally, the rationale for using a qualitative research method.

Chapter 2 presents a review of the recent literature on the topic.

13

Chapter 3 describes the research methodology followed. This is followed by an outline of the process of data collection, how data is analyzed, and the trustworthiness of the study data; and it confirms its compliance with UoT's research ethical procedures.

Chapter 4 presents the results of the case study. The identification of the themes and development of the categories of data is then described and linked with the literature.

Chapter 5 presents the conclusions drawn from the analysis of the data, and recommendations emerging therefrom. This chapter concludes with an outline of the implications of the study for academia and the profession, and suggests further research possibilities.

CHAPTER 2

LITERATURE REVIEW

2.1 Introduction

A literature review was conducted to review recent research in the areas of the role of advisory boards; the characteristics of representation on advisory boards; and key barriers experienced by curriculum developers.

2.2 Role of Advisory Boards

Snead et al., (2019) recognise that accounting advisory boards play a role in commenting on the potential topics and pedagogical approaches to be used in the programme. In Kilcrease's (2011) view, the role of advisory boards is to share experiences that are pertinent to specific academic areas of curriculum and programme development. In sharing their experiences, Conroy et al., (1996) point out that advisory boards provide value through their advice on curriculum matters. Kaupins and Coco (2002) also observed that advisory boards are required to discuss curriculum issues, and to pursue ideas to promote programme accreditation. Zahra et al., (2011) conducted a study on academic advisory boards' contributions to education and learning, reviewing 31 academic advisory boards. The study recommends that specific attention should be given to teaching orientation, content, and the approach adopted by the entrepreneurship programme. Additionally, boards should assist in gauging student learning processes, identifying obstacles to board contributions, and ensuring appropriate emphasis is given to teaching the skills students will need in the profession.

Genheimer and Shehab (2009) also surveyed 90 industry advisory board members responsible for the operation and effectiveness of an engineering education programme. Their study found that effective boards have a clear understanding of their role in influencing curriculum development, motivating for engagement with students, programme accreditation, and encouraging alumni to contribute financial support to the program. Benigni et al., (2011) agree that universities must look externally for resources that will assist them with preparing students for recruitment into employment, and advise programme administrators on how to introduce/include best practices into the curriculum. Arlinghaus (2007) states that advisory boards in accounting have a key role to play in ensuring that programs teach to business and industry standards. For example, Baker et al., (2007) studied the best practices of accounting department advisory boards, by conducting a survey of the chairs of accounting departments. Their findings reveal that in this instance, the accounting curriculum is endorsed and approved by the advisory board. This means that the boards have been given a key role in curriculum review, strategic planning, and providing internship opportunities for students. Baker et al., (2007) believe that advisory boards should be made up of alumni, and Hunt et al., (2017) too, expressed the view that advisory boards are successful when alumni are involved as members. Courtney et al., (2021) state that the composition of advisory boards should include professionals with expertise relevant to the education organization and the industry. Additionally, these professionals should be highly motivated to give their time, energy, and talents to providing coherent advice and support to the faculty (Baker et al., 2007; Hunt et al., 2017).

2.3 Composition of Advisory Boards

According to Boadi and Osarfo (2019), the composition of the board should reflect the diversity within society (involving people of different genders, race, and educational backgrounds), and have a diversity of educational, business and professional qualifications. An advisory board is a representative group of individuals whose experience and abilities represent a cross-section of those required in a particular occupation or profession (Kilcrease, 2011). The purpose of the advisory board is to assist academics in developing and evaluating their academic programs to ensure they serve the needs of students, business, and industry, and to provide expertise and guidance pertaining to technological changes in the field (TUT, 2008). Stensaker et al., (2021) state that advisory boards are established to be consulted on issues of importance to maintaining curriculum relevance. Kolev et al., (2019) mention that advisory boards should include specialist sub-committees responsible for performing critical functions for the advisory boards. Linked to this is the idea that program advisory committees should ensure they provide meaningful contributions to improving the quality of outcomes and employability of graduates (TUT, 2017). Thus, the composition of the advisory boards has the potential to deliver a significantly positive effect on the development of the curriculum, and the employability of the students. Enache and García-Meca (2019) studied the composition of the advisory boards in accounting and recommended that the advisory boards should be made up of outside directors (non-faculty), and appointed for the appropriateness of their skills, abilities, connections, and knowledge in the

industries they are working in. They could equally be drawn from the ranks of business experts, support specialists, and influential/respected community leaders.

2.4 Key barriers faced by curriculum developers and advisory boards

According to Correa Ruiz (2013), it is not enough to simply add new techniques into their professional curricula without enabling accountants to critically examine the ways of thinking and the impact of the techniques that have contributed to society's sustainability. At the UoT, the accounting curriculum, and its application through teaching and learning, is being shaped in accordance with the TUT Transformation Framework 2017: a key focus of this Framework is understanding what it means to be a student, and what role higher education is to play in society (TUT, 2017). Abeysekera (2015) recognizes that in accounting education there are several challenges, not least of which is the decision on which instructional methods to adopt to facilitate student learning. As Bonk and Smith (1998) point out, the consultative model of teaching can meet these challenges.

The accounting curriculum should be built/developed to produce graduates who are betterequipped to enter the job market (Heang et al., (2019). Illustrating this, Dehghani and Pakmehr (2011) studied challenges of curriculum implementation in higher education in interviews with 184 academic managers in the universities of West-Azerbaijan. Their findings revealed that their curriculums did not include managerial competencies, and that the existing curriculum problems faced by these universities did not stem from managerial regulations, but from curriculum implementation. To resolve curriculum challenges (the disconnects with the job market their graduates are about to enter), Correa Ruiz (2013) suggests that the curriculum should also give accountants a comprehensive appreciation of business models, organizational core values, governance risks, and sustainability perspectives. Thus, the accounting curriculum should prepare accounting students for a career in accounting (Dewua & Barghathť, 2019).

In order to facilitate this, advisory boards should be consulted by academic departments and curriculum designers for advice on new developments in accounting, and in the business world in general, so as to ensure their graduates have the skills that guarantee employability. In other words, making use of advisory board members' connections with the fast-changing business sector will facilitate more effective transfer of knowledge of the sophisticated technological skills future accountants will require.

2.5 Summary of the Chapter

The major thrust of this thesis has been to explore faculty perceptions of the value of advisory boards in the development of the accounting education curriculum. Guided by the key research questions, the literature presented in this chapter presents the picture of very deeply immersed researchers who have produced widely researched work about the role of advisory boards; the characteristics that should be represented on advisory boards; and key barriers faced by curriculum developers across multiple universities' accounting curriculums. The literature shows that while there has been much research and discussion conducted on accounting curriculums, and the composition and duties of advisory board committees, there remains a lack of consensus on key elements of such boards' identity. The main purpose of the literature chapter was thus to review the work reflected in previous studies on knowledge sharing with respect to the role of advisory boards/committees on curriculum development.

CHAPTER 3

RESEARCH DESIGN AND METHODOLOGY

3.1 Introduction

This chapter presents the methods and materials used to explore faculty perceptions of the value of advisory boards in developing the accounting education curriculum. The chapter specifically addresses the research paradigm, research design, case study research methodology, population and sampling techniques, data collection and data analysis. This study is based on a qualitative research methodology, in a case study research approach, and is informed by an interpretivism research design/paradigm.

3.2 Research Paradigm

This study uses the interpretivist research paradigm. Interpretivists assume that access to reality happens through social constructions, and that reality is subjective (Alharahsheh & Pius, 2020). According to Nicotera (2019), the interpretivist paradigm assumes that social reality lives internally through human interaction. From the interpretivist point of view, language and personal interpretations play important roles in society, and are essential to interpreting the meaning of data that can be collected around a phenomenon (Curry, 2020; Bevir & Hall, 2020; Jakubik, 2021). Kennedy (2019) states that the interpretivist paradigm fits well with the qualitative research approach. Therefore, it is through the interpretivist paradigm that this study explores faculty perceptions of the value of advisory boards in the accounting education curriculum at the University of Technology.

3.3 Research Methodology

According to Creswell and Creswell (2017), a research methodology provides a strategy by which the researcher can plot out a systemic process to understand a phenomenon. Research methodology simply refers to a research study's strategies and how they are applied practically (Creswell & Creswell, 2017). There are three broad approaches to research, namely: quantitative, qualitative, and mixed methods. This study adopted a qualitative research method. According to Harrison et al., (2020), qualitative research can be characterized as the simultaneous study of many aspects of a phenomenon, and the attempt to study things as they exist naturally. This approach uses inductive reasoning (Pope & Mays, 2020). Kuckartz and Rädiker (2023) acknowledge that inductive reasoning in research requires a logical thinking process that integrates observations with experiential information, and that enables one to draw a conclusion. Therefore, by using a qualitative research strategy, this study explores faculty perceptions of the value of advisory boards in developing the accounting education curriculum at the University of Technology.

An example of a current topic of debate/issue impacting students and future graduates is how to address IFRS. As Clay (2013) suggests in his qualitative analysis of an accounting curriculum, for accounting educators to be effective in addressing current issues in the classroom they need to be knowledgeable with respect to the contents and application of IFRS, and to have determined the best way to include these standards in the accounting curriculum. Similarly, in an earlier study by Larson and Brady (2009), they advise that accounting educators need to plan how to incorporate IFRS into their accounting courses. Van Noy et al., (2012) use a qualitative research approach to reach the conclusion key to curriculum development is the need to achieve programme alignment with the real-world students will shortly enter: thus, effective curriculum development.

3.4 Case Study Research

Case study methodology enables researchers to conduct an in-depth exploration of one or more individuals' programs, events, activities, and processes (Priya, 2021). Yin (2009) describes a case study as an empirical inquiry that investigates a contemporary phenomenon in depth and within its real-life context. Case study research methods are commonly used in social, educational, clinical, and business research (Dul & Hak, 2007; Yin, 2011). A good case study should provide new insights into the subject (Gerring, 2004); propose practical courses of action to resolve a problem (Flyvbjerg, 2006); and/or open new directions for future research (Beverland & Lindgreen, 2010). Making use of a case study research methodology, this study explores the characteristics, meanings, and implications of faculty members' perceptions of the value of advisory boards in developing the accounting education curriculum in the Faculty of Economics and Finance of the University of Technology in South Africa.

3.5 Population of the study

The population of the study was purposively identified, and the sample was selected based on the participants' level of knowledge and understanding of the subject. 'Population' refers to the total of all participants who could give information on the matter that is being reviewed (Babbie, 2020). The population consisted of 41 individuals (listed in Table 3.1) from the faculty of Economics and Finance at the UoT. From this population a sample of 10 participants was invited to participate in the study. These participants were selected from various environments within the faculty, and include the HoDs, academic managers, advisory committee members, and alumni/graduates (see Table 3.1). The rationale for using purposive sampling was that the listed participants met the criteria for the study - were familiar with the curriculum and with the accounting environment. Sampling is a process by which a part of a population is identified as able to represent the whole population (Gill, 2020). Choosing a suitable sample size in qualitative research "is an area of conceptual debate and practical uncertainty" (Vasileiou et al., 2018). For example, Creswell (2002) recommends a minimum sample size of between 3 and 5 members in a case study using qualitative research methods, to reach data saturation, while Stake (2006) recommends researchers conduct between 4 and 10 interviews with participants to achieve data saturation. This researcher purposively selected 10 respondents whose knowledge and responsibilities fitted the outlined criteria for this research study. The criteria were: alignment with the problem statement and the aim of the study, and familiarity with the topics addressed by the research questions. The chosen respondents received invitation letters to participate in the study and consent forms, to be signed and returned with their confirmation of intention to participate.

#	Group of respondents	Population	Sample	Reason for inclusion
1	Heads of Departments (The faculty has 5 academic departments)	5	5	A minimum of 5 respondents was deemed sufficient to provide an understanding of the value of advisory boards in accounting education and the curriculum renewal processes. (This was also the minimum number deemed necessary to get data saturation.)
2	Academic managers	2	1	A minimum of 1 respondent was deemed sufficient. These members are responsible for the inputs into accountancy and curriculum renewal agendas.

Table 3.1: Population and sampling

				These managers both have knowledge of the type of learning content that should be included in the curriculum.
3	Advisory board members	10	2	A minimum of 2 respondent was deemed sufficient. These members are responsible for the board's inputs into the accountancy curriculum revision process.
4	Alumni/graduates	5	2	A minimum of 1 respondent was deemed sufficient to provide input as to the value of advisory boards in accounting education.
	Total number of population and sample size	41	10	

3.6 Data collection technique

The qualitative data collection methods that researchers choose depend on the specific needs of their gualitative inquiry (Creswell & Poth, 2016). Data collection is simply the process of gathering information from all relevant sources prior to finding a solution to the research problem (Gill et al., 2008) and addressing a specific subject. The data collection component of research is common to all fields of study, including physical and social sciences (Mayer, 2015), humanities (Flick, 2018), and business (Adams et al., 2014). According to Guest et al., (2013), data collection methods are simply the techniques and procedures used to gather information for research purposes. Data can be qualitative (meaning contextual in nature), or quantitative (meaning numeric in nature) (Creswell & Poth, 2016; Barrett & Twycross, 2018). An interview is an important qualitative data collection method used for extracting more detailed information about, or to achieve a deep understanding (Rowley, 2012) of the research topic. This study used interviews as the only method to collect data. While recognizing that semi-structured (or unstructured) interviews and/or focus groups are two of the most popular qualitative methods, and that they are usually conducted in face-to-face settings, the face-to-face aspect was not possible as the participants were geographically widely spread and there was insufficient time to visit each participant in person.

3.6.1 e-Interviews

According to Fritz and Vandermause (2018), in-depth e-interviews are meaningful and useful use when respondents are geographically dispersed. Due to the nature of their work and their busy work schedules, using email seemed the most appropriate way to approach respondents if the research was to achieve the desired deeper understanding of faculty perceptions on the value of advisory boards in accounting education curriculum development. Thus, qualitative e-interview was the preferred method to collect primary data for this research.

In qualitative methodology the e-interview is usually seen as being flexible (Bampton et al., 2013; Bauman, 2015). Online interviewing is another qualitative research method (Olivero & Lunt, 2004), "using a wide range of synchronous and asynchronous technologies for interviews" (Janghorban et al., 2014). Busher and James (2009) see electronic interviews as a form of data collection which continues to gain trust with interviewees, and is thus as an important data collection option. Cotton (2015) states that, if it is not possible to do interviews in person, it is strongly recommended to do e-interviews to collect data. The use of email to conduct interviews is thought to be an effective way to collect qualitative data (Fritz & Vandermause, 2018), and it is also recognized as an effective cost saving method (Hawkins, 2018).Given that interviews can be conducted in a variety of different forms (for example, by telephone or online methods), it was the researcher's choice to conduct interviews via email, as opposed to conducting face-to-face interviews.

There are different approaches to qualitative interviews – they can be unstructured, semistructured and formally structured in design. With the unstructured interview, the researcher might start the conversation with a question, while the semi-structured interview allows the researcher to follow a checklist of topics and questions that need to be covered, and guide the conversation accordingly. The aim of this study is to explore perceptions on the value of advisory boards to the faculties involved in accounting education curriculum development. Thus, this research benefits from the use of a qualitative study approach using semi-structured interviews, as this allowed flexibility in responses and data collection. This method provided the researcher with the opportunity to ask respondents (HoDs, subject coordinators, advisory committee members, and alumni/graduates) about the topic, and to pose certain probing questions related to the topic, for amplification of their responses. The semi-structured interviews also allowed the researcher to

23

manage the order of the questions (when additional clarification was required) and to ask the probing questions. The semi-structured interviews were guided by the following key questions.

3.6.2 Primary research questions

The research questions were organized to explore three main areas:

Role of advisory boards

• RQ1: How do participants perceive the role of advisory boards in the accounting curriculum development?

Composition of advisory boards

• RQ2: What is the composition of advisory boards in the faculty?

Challenges of the curriculum in the faulty

• RQ3: What are the challenges of the curriculum development in the faculty?

For detailed interview procedures see Annexure 1, which contains the interview questions used for the data collection. The interview questions were emailed to all 10 of the intended respondents, and their responses were received within eight (8) days. A 100% response rate was achieved.

3.7 Data analysis technique

The main purpose of data analysis is to draw conclusions from specific data (Wickham & Wickham, 2016; Lemon & Hayes, 2020), and to interpret these conclusions/results (Adu, 2019; Linneberg & Korsgaard, 2019). Qualitative data analysis involves labor-intensive, in-depth processes (Adu, 2019; Lester et al., 2020; Lemon & Hayes, 2020). It deals with data from interview transcripts, notes, and text documents (Wickham & Wickham, 2016; Lemon & Hayes, 2020). Qualitative data analysis requires data preparation and arrangement (Adu, 2019), review and exploration of data (Lester et al., 2020), and creation of codes that are subsequently linked to specific themes (Linneberg and Korsgaard, 2019), and finally, the presentation of the themes in a cohesive narrative manner (Lochmiller, 2021). Qualitative data analysis is inductive, and focuses on meaning (Bingham & Witkowsky, 2021). Hwang (2008) states that utilizing qualitative data analysis software (such as Atlas.ti) is important and useful, but is not a prerequisite for

undertaking qualitative analysis. There is a range of programs available that assist the qualitative researcher in analyzing qualitative data, including Qualitative Data Analysis Software (QDAS) and NVivo, in addition to Atlas.ti. This type of software supports the coding of data and the identification of themes and patterns. Bazeley (2009) mentions that the identification of significant themes and patterns requires skill if qualitative data analysis is to be successfully used. One way of analyzing data is to sort it (Ruona, 2005). Thereafter, theme identification is one of the most fundamental tasks in qualitative research (Ryan & Bernard, 2003).

3.7.1 Data Coding

According to Elliott (2018), coding is a way of indexing or mapping data to provide an overview of different categories/classes of data: this then allows the researcher to make sense of the data in relation to the research questions. Basit (2003) adds that coding is one of the significant steps to organize and make sense of textual data, and it can be done manually or electronically. Manual coding requires researchers to read through all the data and to develop and manually assign codes and themes to ideas and phrases (Wicks, 2017; Saldaña, 2021). The researcher used manual coding in this research, analyzing the response text line-by-line, inserting comments as appropriate. This process allowed the researcher to identify the important/significant lines and to make notes regarding the text data he would likely be quoting in this report. Saldaña (2021) is of the view that data coding can be guided by the research questions, by the literature review, and the responses of the participants.

3.7.2 Coding and analysis

It is noteworthy to remark that the categories need to be determined beforehand in order to perform the data analysis and coding. Table 4.2 identifies and explains the categories, as an initial step. After reading the participants' comments, the researcher marked any that initially appeared to be pertinent. The themes were developed by quoting the statements and associated participant responses, and listing them in order of similarity. Certain statements were taken verbatim, while others were summarized. The statements were examined and analyzed, after the coding of each participant's responses (see Chapter 4 of the study).

3.8 Trustworthiness

There are four concepts used in qualitative research to measure the trustworthiness of the study results. According to Korstjens and Moser (2018), these concepts include credibility, transferability, dependability, and confirmability, and are explained below:

- Credibility The credibility of qualitative data can be assured through multiple reviews throughout data collection process to ensure data is appropriate. The credibility in this study can be established by checking preliminary findings and interpretations against the raw data. Further, validation and checking of participants' responses can be done through Table 3.1 of the population.
- Transferability Transferability is similar to generalizability in quantitative research methodology; however, it is not generalizability. In this study, the transferability can be measured through the description of the findings, for anyone who would like to transfer the findings can do so.
- Dependability Dependability can be ensured through creation of a clear audit trail indicating the rigorous data collection techniques and procedures used, and ensuring the analysis is coherent and well documented. Typically, an inquiry audit using an outside reviewer assures dependability. This study realized dependability through reliance on ethical issues such as the researcher's responsibility, and trust.
- Confirmability Confirmability of qualitative data is assured when data are checked and rechecked throughout the data collection and analysis processes, to ensure findings would likely be repeatable by subsequent researchers. Therefore, confirmability can be achieved and reassured through use of verbatim statements from participants, which thus avoids the potential of bias and manipulation.

3.9 Ethical consideration

Ethical considerations are integral to research and need to remain at the forefront of every study (Khan, 2014). According to Sieber and Tolich (2012), responsible researchers need to think about ethics when preparing their research proposals and throughout the subsequent research projects.

Ethics in research guides the protection of participants' rights and their well-being (Arifin, 2018). Compliance with accepted research ethics ensures that findings are valid and reliable (Campbell & Cowton, 2015). This is not simply a tick-box compliance issue. It is a crucial consideration that can make all the difference to the outcome of a research project. Implementing ethical guidelines is essential (Kosinski et al., 2015; Kent et al., 2002). The Haaga Helia University of Applied Sciences and the University of Technology require every researcher to adhere to the fundamental research ethics principles of low risk of harm, obtaining informed consent, and protecting the anonymity and confidentiality of participants and others. This study has been conducted in rigorous compliance with the HEDS-RICH-REC and Research and Ethics Committee requirements, as best practices designed to protect participants' rights and well-being. The research proposal, methodology and research questions were all thoroughly reviewed to ensure there were no potential risks or unmerited benefits. This was done before the start of the data collection and completion phases of the study. In terms of the validity of the research and research design, the Research and Ethics Committee requested clarity on how specific research questions were to be used. This was given and satisfactorily received. Thus, the ethics committee ensured that the methods used related specifically to obtaining responses to the research questions.

3.10 Summary of the Chapter

In this chapter the research methodology used in this study was discussed. The use of a qualitative research method for data collection and analysis was explained and justified. The research paradigm, case study, population, ethical procedures, and measures followed during the data collection were also discussed, and the information about the sample was provided.

CHAPTER 4

PRESENTATION OF RESULTS

4.1 Introduction

The findings of this study are reported next, as part of the process to explore faculty perceptions of the value of advisory boards in accounting education curriculum development. Findings are then analysed in respect of each of the research questions. Thus: How do participants perceive the role of advisory boards in accounting curriculum development? What is the composition of faculty advisory boards? What are the challenges faced by curriculum development in the faculty? These categories were used to develop the themes.

4.2 Section A: Profile of the Participants

The questions used to guide the interviews were specially created with this study in mind. To gain an understanding of the participants' backgrounds, **Table 4.1** presents the demographic characteristics of the participants. Specific questions were included in the emailed questionnaire to gather data regarding the participants' profiles (see Annexure 1). (The research questions were developed following the reading and discussion of the pertinent literature.) However, due to ethical concerns, some participant information was reported under pseudonyms. A total of ten (10) respondents (academic managers, graduates, and heads of department) participated in the study. These participants are informed individuals, fully aware of the importance of advisory boards in accounting education, and have extensive prior experience with them. They have all participated in the university's academic advisory committee system.

Table 4.1 displays the demographic information for each of the 10 participants in the study: current employment position; number of years in the present position and in their previous positions, and highest academic qualification attained. Studies conducted by Zimmerman et al., (2000); Baker et al., (2007); Norman and Bagranoff (2019), and Xu et al., (2023) have shown that demographic information such as administrative/academic position of the participant, duration in the position, previous position held, and highest academic qualification is important to understanding the responses of the participants.

Participants	Position of the	Number of years	The previous	Highest academic
number	participant	in the position	position held	qualifications
1	HoD	3 years	9 years	Master's degree
2	HoD	6 years	8 years	Master's degree
3	HoD	14 years	22 years	Doctoral degree
4	HoD	1 year	7 years	Master's degree
5	HoD	4 years	10 years	Master's degree
6	Academic manager	10 years	12 years	Master's degree
7	Academic manager	2 years	5 years	Honour's degree
8	Alumni/graduates	3 years	2 years	Honour's degree
9	Alumni/graduates	6 years	4 years	BTech degree
10	Alumni/graduates	12 years	8 years	BTech degree

 Table 4.1: Demographic information of the participants

The data collection guidelines were used so as to obtain standardized information from the participants. **Table 4.1** reports the demographic information of participants in the study that explored faculty perceptions of the value of advisory boards in developing the accounting education curriculum.

4.3 Thematic Analysis

As part of the qualitative investigation, ten participants were sent interview questions in a single email. Every response to the interview questions underwent human coding, categorization, and analysis (i.e., no software was used to process the raw data). All interviews and analyses were completed in the month of September 2023. Codes and categories were used to build themes. Out of the ten interview questions, eight themes were identified. These were derived from the participants' responses.

Interview question	How do participants perceive the role of
	advisory boards in accounting curriculum
	development?
Categories - Role of advisory boa	rds
Theme 1	Theme 2

Table 4.2: Themes and categories corresponding to interview questions

Interview question	What is the composition of advisory boards in the faculty?			
Categories - Composition of advisory boards				
Theme 1	Theme 1Theme 2Theme 3			
Membership	Graduates	Industry Experts		

Interview question	What are the challenges curriculum development faces in the faculty?				
Categories - Challenges of the curriculum in the faculty					
Theme 1	Theme 2 Theme 3				
Up-to-date curriculum	Professional bodies	International recognition			

RQ1 – How do participants perceive the role of advisory boards in the accounting curriculum?

Participants' responses to the question on how they perceive the role of advisory boards in accounting curriculum development are discussed next.

4.3.1 Category – Role of advisory boards

As a point of departure, Tapis and Delaney (2017) state that using accounting advisory boards is a way to promote collaboration between faculty and practitioners. The role of the academic advisory boards is thus to provide guidance and advice to academics on how to develop, continuously improve, and maintain the relevance of an accounting curriculum.

Seven of the 10 study participants (participants #1, 2, 3, 4, 5, 6, and 7) stated that they were generally satisfied with the role of advisory boards in their departments. Three of the participants

(#8, 9, and 10) additionally agreed that the advisory committees were able to assist in generating career opportunities for graduates or for academics wanting to move into commerce/industry.

The participants were articulate/knowledgeable on the role of advisory boards in the faculty, and on their importance to the faculty and the university. A composite summary of participant responses follows:

"The role of the Academic Advisory Committee (AAC) is to advise academic departments on matters of the curriculum (participants #1 and 3); career opportunities that exist for student (participant #2) and [to maintain...] the quality of the academic offering (participants #4 and 6). Since the AAC is composed of industry experts and alumni, the faculty and by extension, the university [the faculty] gains a good reputation among industry players, resulting in employers having confidence that the faculty produces competent graduates". (participants # 5 and 7).

To understand how participants perceived the role of advisory boards in the accounting curriculum development process, participants' responses were unanimous, showing the level of their perceptions about the role of advisory boards in the faculty. A composite summary of 3 participants responses:

"From my understanding, the formulation of the advisory board best supports the faculty or the department to succeed in its qualification(s) offering, and professional development pathways (participant #4). They are mentoring the faculty academic staff to adapt to transitional values, ethics and code of conduct (participant #5). They also assist the faculty in exploring new opportunities and stimulating high-quality conversation, and of course, they provide expert recommendations". (participant #1).

Specific issues that advisory boards in the faculty should address/are addressing were identified. All participants (participants #1, 2, 3, 4, 5, 6, and 7, and #8, 9 and I0) made use of the opportunity to identify tasks the academic advisory boards could perform to benefit the departments are indicated below in a representative quote:

"Stimulate critical thinking, analysis, and heighten decision-makers' awareness (participant #1). Provide expert recommendation (participant #2). Help with value-driven insights on qualification(s) to evolve with the relevant new trends (participant #3). Benchmarking of the qualification(s) with national and international standards (participant #4). Evaluation of effectiveness and best practices (participant #5). Mentoring the faculty/department academic staff to adapt to transitional values, ethics and code of conduct (participant #6). Help with value-driven insights on qualification(s) to evolve with the relevant new trends (participant #7). Explore new opportunities and stimulate high-quality conversation (participant #8). Stimulate critical thinking, and analysis, and heighten decision-makers' awareness (participant #9). Stipulate expert recommendation" (participant #10).

Theme 1 – Support

Xu et al., (2023) report on the effectiveness of advisory boards in accounting programs, and conclude that the most effective activities undertaken by accounting advisory boards are in the area of student career development. The academic advisory committee in accounting thus is seen as serving as a bridge between the faculty and the business community, including government institutions. The support from advisory boards (the advice and guidance they offer (Hepperlen et al., 2022), and the collaboration they facilitate, enhances the quality of the mission of the faculty (Guimón, 2013).

Participants were also interested in expressing their views about the importance of the role of advisory boards:

"The support is certainly helpful and certainly implementable. For instance, there are regulations that prescribe how preferential procurement in Government should be dealt with. When the Constitution declared parts of the regulations as unconstitutional, it created a challenge for us because we teach these regulations. So, members of the AAC assisted us with documents of procedures that they followed until the new procurement bill was implemented (participant #1). In my opinion, the responsibility lies with the department to ensure that they create conditions for AAC to effectively exercise their role (participant #2). The faculty plays a supporting role and will support what the department proposes" – (participant #8).

This illustrates effective collaboration between the faculty and the business community and the support that should ensure that accounting graduates have good professional opportunities available upon graduation.

Theme 2 – Curriculum

The curriculum is explained as a prearranged course of learning set for students (Suskie, 2018). It tells what, why, and how students should learn. It is meant to help students develop the skills,

knowledge, values, and attitudes they need to be successful both inside and outside of the university.

The accounting curriculum is a selected specialization within the accountancy programme. There are specialized emphasis areas like financial accounting, internal auditing, public sector finance, finance and investment, and economics under the broad umbrella of "accounting". The coverage of the curriculum's core courses often includes a set of basic business subjects, such as business management, communication for academic purposes, business research, taxation, financial accounting, auditing, finance, and economics.

There is a general notion that advisory boards are useful in the accounting curriculum and most of the respondents tend to support the idea that ... "the advisory boards' role is to guide the curriculum and provide review of the current program offerings" (participant #1). Furthermore, it is through the advisory boards that the institution will offer relevant qualifications needed by industry and the upliftment of the society" – (participant #4). Participants have also confirmed that "the advisory board meetings are held two times per academic year (participant #5). Each academic department has its advisory board related to the academic qualifications offered in the department" – (participant #7).

Curriculum development is regarded as catch-all phrase for a continuing process in which structure and systematic planning processes are initiated, developed, adopted, and evaluated (Bengio et al., 2009). In the context of an accounting curriculum, it should give students an understanding of basic business and accounting practices, and it can also prepare students to move forward into professional careers as accountants, accounting technicians, and business entrepreneurs. Thus, the guidance from the advisory boards assists the academic departments in preparing their students for careers in entrepreneurship, or to take positions in government, industry, corporate enterprises and accounting practices. For this, though, they need appropriate skills. These observations from recent research resonate with the findings of the study by Maali and Al-Attar (2020) who examined the accounting curricula in universities in Jordanian universities and the employment market needs. Their study identified the skills and competencies that businesses require from accounting graduates. The study also found that quality assurance bodies are strangling universities' freedom to develop curriculums that meet the market's needs: statutory oversight bodies are preventing curriculum development that would align courses with what potential employers need. On the other hand, Snead et al., (2019) also point out that the use of advisory boards' input for revising the accountancy curriculum continues to be important to ensure the topics covered in the course programme are those that commerce and industry need.

Summary – What should be the role of advisory board?

The academic advisory committee's function in accounting education should be that of a resource for information and provide support for developing currently relevant curricula. The advisory committee should support the faculty's active involvement in academic research, facilitate interactions with the professions, and assist in other areas that are essential to maintaining and growing a vibrant accounting program. Additionally, by fostering a favorable perception of the accounting program among professionals, business leaders, and the public, the committee should thus serve as a conduit for communication between the faculty and the professional accountancy community.

RQ2 – What is the composition of advisory boards in the faculty?

The participants were asked about the composition of the advisory boards in the faculty: the themes the participants' responses supported are summarized next:

4.3.2 Category – Composition of advisory boards

Theme 1 – Membership of advisory boards

".... our members are from industry, government, academia, and professional bodies (I am referring to the external people, by the way ...)." (participant #1). "Additional people are lecturers in the department" – (participant #3).

The participants recognised that academic advisory boards bring external perceptions to the faculty, since they include various external (non-academic) members from industry, government, and professional bodies.

Drawing from the responses it can be concluded that committee members' inputs bring the following benefits to the departments:

• knowledge resources;

- information about new developments in the marketplace;
- curriculum discussions in current business/economic contexts;
- an ability to create jobs and placement opportunities for students;
- the facilitation of industry partnerships, and signing of memorandums of understanding between industry leaders and the departments; and
- creating opportunities to present guest lectures by industry leaders and external committee members.

Thus, as the views of participant #2 strongly recommends, advisory board members can ".... increase their involvement in the programme through, for instance, using them as guest lecturers. This will ensure that they interact continuously with the department or faculty".

Theme 2 – Graduates

The success of graduates/alumni connects the Faculty of Economics and Finance with industry and government. According to Jackson et al., (2022), this is an important means to attract accounting and finance graduates to advisory boards.

Participants (participants #1, 2, 3, 4, 5, 6, 7, 8, 9, and 10) identified the involvement of UoT graduates on the advisory boards as something that is needed: "Yes, we do have one, a former student who works at the National Treasury as a Local Government Analyst (participant #1). In my opinion, we should have at least three alumni from different sectors, who will then be able to advocate for the quality of our programs in their organizations" (participant #2).

Some participants indicated that "... we have a departmental alumni network which keeps us (participant #3) "... in the loop with old students ..." (participant #4). Also, try to understand what they are doing in life (participant #5) "... we sometimes even try to communicate and keep in touch with them" (participant #6). "Although they are busy, we also invite them to participate or just to attend departmental events and to share their experiences with students – very important" (participants #7 and 8). "I would of course, like graduates to raise bursary ..." (participant #9) "... and scholarship opportunities for students..." (participant #10).

Theme 3 – Industry Experts

The academic advisory boards in the accounting departments provide the insight and direction that helps the Faculty of Economics and Finance to stay familiar with current professional developments and aware of industry needs (Ho, 2023). The industry experts in the advisory committees ensure that the faculty programme meets the needs of the accounting profession (Edeigba, 2022; Mousavi et al., 2022). Most of the participants (participants #2, 3, 4, 6, 7, 9, and 10) confirmed that the composition of advisory boards in the faculty comprises an "External Independent Chairperson (participant #2); External members attached to the Community of Best Practice (participant #3); external members from Professional Bodies (participant #4); and internal members of the academic staff" (participant #6). On the recruitment of the advisory committee, all the participants who took part in the data collection session indicated that: "the composition of the Academic Advisory Committee is in line with the policy on Academic Advisory Committees which requires a mixture of industry experts, alumni, a final year student and internal staff members who are in the service of the relevant academic department (participant #7). "Members who have an influential standing in their field are recruited to the ACC, as this assists the department in ensuring that advice received represents up-to-date industry practices" (participant #9) ".... again, staff, students, and the Institution benefit from this ..." (participant #10).

Summary – Who should be on the advisory board committees?

People with expertise in the subject of accountancy should make up the academic advisory board for the accounting department. To facilitate the sharing of information across curricula, such membership should include those who are currently employed by other universities in addition to those who are actively practicing as professional accountants. The group of people who comprise the advisory committee membership also includes industry experts and alumni/graduates who oversee professional accountants. These individuals may play a big role in advising academic departments about emerging trends and career advancement opportunities.

RQ 3 – What are the challenges of the curriculum development in the faculty?

When the participants were asked about what the challenges the curriculum faced in the faculty representative participant responses included the following:

4.3.3 Category – Challenges of the curriculum development in the faulty

Theme 1 – Up-to-date curriculum

Academic advisory boards play a significant role in accounting and finance curricula (Senaratne et al., 2022) to ensure that the accounting curriculum is up-to-date and relevant, and meets the market's needs (Twyford & Dean, 2023). An up-to-date curriculum offers the opportunity to teach relevant content that addresses modern business and industry needs (Griffin, 2022), thereby equipping students with the skills and experience sought by employers (Jackson, 2023; Heng & Sol, 2023). This is confirmed by the following participants' comment: (participants #2 and #3) "that an up-to-date curriculum should be measured on the amendment of [texts to teach] relevant new trends within the field of accounting (participant #2) ".... and be implemented in the workbooks, classroom, and assessments" (participant #3).

According to Samanthi and Gooneratne (2023), an up-to-date curriculum empowers students to gain appropriate expertise in areas including financial and management accounting, corporate finance, accounting and finance issues, and business technology. All et al., (2022) advocate transforming assessment in accounting education to align with online learning approaches.

Theme 2 – Professional bodies

The professional bodies assist the faculty by accrediting their courses and certifying that they meet the high standards of the professional accountancy bodies such as the South African Institute of Professional Accountants (SAIPA), the Association of Chartered Certified Accountants (ACCA), the Chartered Institute of Management Accountants (CIMA), the South African Institute of Taxation (SAIT), the Institute of Internal Auditors (IIA), the Chartered Institute of Government Finance Audit and Risk Officers (CIGFARO), and the Southern African Institute for Business Accountants (SAIBA).

On the question of whether there are challenges to the curriculum in the faculty - the results of the findings are interesting. Participant #6 expressed the following views:

"... achieving some of the simulated work integrated learning outcomes [is difficult] due to availability of buildings... Attracting industry specialists for seminars [is also difficult] due to the perceptions some have regarding the student unrests and the limited academic period [that has been] interrupted by students strikes".

Thus, it is believed that programme accreditation by the professional bodies will fill curriculum gaps (and increase students' learning with exposure to real-world experiences). Students who have completed an accredited degree are in high demand by employers. Thus, the role of academic advisory boards can be extended to benefit students and the faculty by advocating for accreditation with professional bodies, and by promoting students as desirable employees because the courses they have completed are relevant now. Thus, the academic advisory committee should actively support and encourage professional bodies to create opportunities to improve graduates' employability (Trung, 2016), and to encourage undergraduates and graduates to look for work in accountancy practices to gain relevant skills and experience.

Theme 3 – International Recognition

The international recognition of academic qualification is achieved through affiliation to international professional bodies, and by adhering to current international accounting standards. Assisting teaching staff to achieve similar affiliations to and membership of international bodies, and having their academic and research credentials recognised improves the status of the courses they teach and recognises they are in good standing with the professions. The following comment articulates the participants' view that *"we can support the faculty to build an effective and efficient international status …. through our networks"* – (participant #7).

Although the faculty may have provided the international community with graduates who possess a global outlook, and with conceptual and applied knowledge in the fields of accounting and finance that will enable them to make a positive impact on society, as participant #7 observed: there are challenges and gaps that still need to be addressed in the accounting curriculum. These involve the "evolution of technology, Artificial Intelligence (AI), and automation in accounting...."

The modules therefore need to be broad and to cover key topics that address current information technology perspectives. Some participants (#7 and #8) feel that the faculty is still experiencing challenges in this area, and observed:

"International collaboration and partnership with other Universities outside South Africa [is not being actively promoted]; (participant #7) And [I] also feel that [the] university is not 'contributing meaningfully at the international forums" (participant #8).

Another challenge recognised by a respondent is that the faculty is not fully utilizing the expertise of individual members of their advisory boards, by not using their guest lecturing potential. There are other areas that are equally challenging and relevant to advisory boards. Thus: *"We should, as faculties, work hard to show members the important role they play in the faculty, especially given the fact that they do this voluntarily. At each meeting, demonstrate to them how their advice was implemented, and utilize them as guest Lecturers to share their industry experience" (Participant #7)*

Participants' presentations of their views about advisory boards embodied the study's . Thus, the HoD expression of the role of advisory boards in the accounting curriculum provides an account of the expectations.

Summary – What is the challenge?

One of the difficulties faced by educators is finding advisory board members who can open doors for our graduates to find full-time jobs. In comparison with other traditional training colleges and universities, the UoT's Faculty finds it difficult to attract suitably qualified applicants with significant industry influence to fill open positions on their departmental advisory boards. Consequently, the faculty is finding that it only attracts recent graduates and alumni, who have not yet gained much accounting experience. Additional difficulties are research-related, and associated issues including curriculum development and embracing technological advances, issues that are likely to guarantee the growth and continuation of an engaging accounting curriculum.

4.4 Summary of the chapter

This chapter presented the results that emerged from the perceptions of the value of advisory boards to the accounting education curriculum and shared by the 10 responders to the in-depth interviews of HoDs, academic managers, advisory board members, and graduates. Through

thematic analysis, themes were identified and linked to the research questions. These themes included support; curriculum development; membership of professional associations and bodies; graduates; industry experts; and international recognition of degrees and research efforts.

In the fifth and final chapter a conclusion and recommendations are provided, an overview of the entire study is outlined, limitations of the study are recognised and recommendations for future research are presented.

CHAPTER 5

CONCLUSION AND RECOMMENDATIONS

5.1 Introduction

This chapter contains the conclusions drawn from the analysis of the data, and recommendations arising from the research study results. The study aimed to explore faculty perceptions of the value of advisory boards in the development of the accounting education curriculum. The study does so by means of semi-structured e-interviews conducted with 10 respondents in which their views were invited regarding the role of advisory boards in accounting curriculum development; the composition of advisory boards; and key barriers to curriculum developers and advisory boards.

The chapter begins with an overview/summary of this research study, and this is followed by the main conclusions drawn from the results and data analysis presented in Chapter 4. The chapter then provides a discussion of the limitations of the study, and outlines the research contributions and implications of the study. A future research direction in the value of advisory boards in the accounting education curriculum is also discussed. Finally, the chapter concludes with a summary, final conclusion and recommendations.

5.2 Overview of the study

This chapter presents a coherent summary of the role of advisory boards in accounting curriculum development as seen by the 10 respondents; discusses the composition of advisory boards; and identifies key barriers experienced by curriculum developers and advisory boards.

A recurring theme in all the preceding chapters is that there are many different schools of thought on this process of curriculum development. In summary, therefore, Chapter 1 presented the definition of the research problem, and outlined the research objectives driving the research that underpins this mini thesis. Thus, Chapter 1 provided detailed information on the motivation for the study. The study focused on the following objectives:

- to determine the role of advisory boards in accounting curriculum development (Study objective 1)
- to establish the composition of advisory boards (*Study objective 2*)
- to identify key barriers of curriculum developers and advisory boards (Study objective 3)

To achieve these three research objectives, the researcher first reviewed the literature (see Chapter 2) on the role of advisory boards in accounting curriculum development; the composition of advisory boards; and key barriers to curriculum developers and advisory boards. The chapter further identified, discussed, and justified the role played by the advisory boards in accounting curriculum development within the University of Technology. Thus, the study's three research objectives were defined, researched and addressed by analysis of the responses to the e-interviews (see Chapters 2 and 4).

5.3 Conclusions drawn from the data analysis

The research methodology and processes adopted to achieve the study's research aim are described in Chapter 3. The views quoted from the literature review (presented in Chapter 2) were largely supported by the empirical results. This chapter also identified that the target population for this study were heads of departments, academic managers, advisory board, and graduates of the faculty. Data was collected via an e-interview in which questions were emailed directly to all participants. Completed responses were returned through the same channels. Data collection and processing methodology was discussed in detail in Chapter 3, and Chapter 4 presented the findings obtained from the previously coded and analysed data.

This qualitative study, employing a case study methodology, provided answers to the research questions. The findings in response to RQ1 suggest that advisory board committees do have the role to play in curriculum support (including efforts to remain abreast of latest industry developments), while participants responses to RQ2 suggest that the evaluation of membership criteria for advisory boards should be reconsidered so as to include more graduates and industry experts. Furthermore, the findings in response to RQ3 also suggest that obtaining accreditation of the curriculum from professional bodies remains a challenge to the advisory board: professional bodies are still reluctant to fully accredit the academic departments' programs. The following

section provides an overview of the empirical results emerging from analyzing the responses to the research questions.

RQ #1: How do participants perceive the role of advisory boards in accounting curriculum development?

Curriculum and Support

Based on an analysis of the data gathered, the advisory board/committee should assess textbooks and other instructional materials to ensure the content is current (curriculum and support). Advisory boards should also assist academic departments to find funding so their teachers may attend conferences and professional gatherings. Educational establishments are aware of the issues and aim to enhance their advisory boards in order to provide more effective curriculum support (McPeak, 2012; Gabbin et al., 2020). According to Kilcrease (2011), advisory boards/committees should also assist with research, teaching and learning, and promote community involvement by motivating for new curricula (to address related academic needs), and promote program enhancements.

RQ #2: What is the composition of advisory boards in the faculty?

Membership

The study concludes that the membership of advisory committees should be opened to anyone who is willing to participate. The recruitment should be done (when seats are available) through announcements in local media and on professional platforms, and the university's website. The study further suggests that the curriculum advisor should be a member of the advisory board/committee. The study also suggests that the advisory boards should include experts in financial accounting, auditing, management reporting, information systems, cost accounting, and taxation. This spread of skills and specialisations will help infuse the course with new ideas, and enable effective review of brainstorming sessions on matters concerning the curriculum. On the other hand, the view was also provided that the composition of the academic advisory boards should be carefully curated. In other words, the study suggests that membership must be balanced, to ensure a high level of expertise and experience (Hunt et al., 2017), and to ensure

appropriate balances between members in public and private practice, and between alumni and non-alumni.

Graduates

The advisory board/committee members should be graduates, to ensure a high academic standard. Jackson et al., (2022) state that the graduates' role on the committee should be to make recommendations on curriculum improvement that will promote students' employability. Additionally, the participation of graduates on the advisory board will increase the visibility of the faculty's programs amongst their employers and professional associates, thereby promoting the program to leaders in business and government. However, the use of the institution's alumni of should not be a prerequisite for participating on an advisory board. The key issue is that it is important to have a variety of skills and experiences from various institutions as this is likely to expand the number and variety of ideas and viewpoints presented at advisory board meetings.

Industry experts

It is vital for the advisory board members to work closely with industry experts in delivering on its mandate of assisting to improve teaching and learning. Having industry experts on the advisory board is also crucial as they provide connections between the university and industry (Edeigba, 2022; Mousavi et al., 2022). Thus, the accounting advisory council brings expertise to the academic department, the university and to the classroom. In particular, here at UoT, the advisory boards work with HoDs by advising and sharing curriculum and career information. Having the industry experts advising the faculty on the latest trends in the accounting profession keeps the faculty and the university on target to ensure the students' skills are aligned with market needs. Having industry experts on the advisory boards is therefore integral to the faculty's connection to the business community and the employment market for their graduates.

RQ #3: What are the challenges of the curriculum development in the faculty?

Professional bodies' recognition

The faculty works with professional bodies like the South African Institute of Professional Accountants (SAIPA), the Association of Chartered Certified Accountants (ACCA), the Chartered

Institute of Management Accountants (CIMA), the South African Institute of Taxation (SAIT), the Institute of Internal Auditors (IIA), the Chartered Institute of Government Finance Audit and Risk Officers (CIGFARO), and the Southern African Institute for Business Accountants (SAIBA) to ensure that qualifications are fit for purpose and meet industry standards. These professional bodies accept/endorse the university's qualifications in a number of ways, such as by recognizing them as a route to employment, and also by providing exemptions for professional membership. Thus, the study recommends that professional bodies be invited to send representatives to participate in the advisory boards as this would cultivate important relationships likely to ensure that accounting students can be confident that they are being taught the practical skills needed to be work-ready upon graduation, and that there will be support for them with career progression and professional recognition.

According to Trung (2016), professional bodies are able to 'brand' (endorse) academic programmes so that they are recognized by professional groups. In line with this, the study participants also recommend that members of professional associations in the business world also be invited to become advisory board members. In other words, by establishing working relationships with professional and business associations, the university departments and faculty can be more effectively guided when updating and revising the curriculum and associated teaching methods.

5.4 Implication of the study

That the research problem was effectively addressed is confirmed by the analysis of the qualitative data. It was demonstrated that advisory boards/committees can directly support curriculum development, help graduates find work, and advise academic departments on how to create curricula that match worldwide standards and are accepted by professional associations. The study's findings should therefore have an impact on advisory boards' involvement in accounting curriculum development. The study's conclusions suggest that there are two things that the accounting field's academic program and the research community should be aware of: Firstly, there may be a chance to investigate what is presumably a new area for academic advisory committees. Even if academic advisory committees are nominally part of UoT, their function should be limited to enhancing/maintaining current relevance of the curriculum, focusing on different aspects of technology as applicable to the department's efforts to teach relevant skills, and fostering strong connections between the academic, business, and professional groups. According to the study, the academic advisory board is crucial in helping to build links and ties

45

between the UoT, academic departments within the faculty, and between the students and their likely employers. Therefore, it follows that their duties would include making sure that the academic credentials on offer match the requirements of professional bodies, and serving as a helpful source of knowledge about ongoing developments in the accounting industry.

The qualifications that graduates attain should be robust enough to gain international recognition in the twenty-first century, as proof that advisory boards have contributed positively to curriculum development and support. The research's usefulness lies in the identification of advisory committees' perceived ability to identify specific content shortcomings in the curriculum, and to address the misalignment of certain skills currently being taught, and those already being demanded by industry. In other words, this research will help the advisory board to identify industry trends, and to offer advice on the academic initiatives' strategic direction.

5.5 Limitations of the study

This study explored predominantly faculty perceptions of the value of advisory boards in the development of the accounting education curriculum. The study used qualitative research methods to collect data from a small sample comprising heads of academic departments, academic managers, and graduates of the faculty. Every study has limitations, and because the methodology used was exclusively qualitative, and made use of emailed e-interviews with probing questions (but without the option of subsequent interrogation of unclear responses) it was not possible to fully quantify the results. Furthermore, the findings of the study are not automatically applicable to all university faculties at UoT. These limitations were considered when the findings were discussed.

5.6 Recommendations and future research

As this thesis focused on the exploration of perceptions of the value of advisory boards in the accounting education curriculum, the academic research and literature context presented in this study might have been overly narrow in its focus. Future research might benefit from being conducted in a wider, even global context. Nevertheless, this research, in attempting to address these study objectives (to determine the role of advisory boards in accounting curriculum development - (Study objective 1): to establish the composition of advisory boards - (Study objective 2): to identify key barriers of curriculum developers and advisory boards - (Study

objective 3) does provide a solid foundation for future local research endeavours. Thus, the following recommendations for future studies are offered:

- More research still needs to be conducted on the relevance of the accounting curriculum in higher education.
- The alignment of the accounting curriculum with the International Standards needs to be investigated, with the research being directed towards new developments and new Standards.
- Given that the curriculum is ever-changing particularly in respect of using and accommodating technological and IT developments, there is a need to do research into efforts to keep up with these technological trends.

5.7 Final conclusion of the study

The main function of advisory boards should be to provide advice to the faculty on professional, teaching, learning, research, and community engagement issues. While noting the contributions and limitations of this study, it is worthwhile, in conclusion, to again highlight that this study has confirmed that the involvement of graduates/alumni on the advisory boards is important.

Lastly, the study concluded with an overview of this thesis and a presentation of a formal conclusion arising from a discussion of the research presented.

REFERENCES

Abeysekera, I. (2015). Student preferences for instructional methods in an accounting curriculum. *International Journal of Teaching and Learning in Higher Education*, *27*(3), 310-319.

Adams, J., Khan, H. T., & Raeside, R. (2014). *Research methods for business and social science students*. SAGE Publications India.

Adu, P. (2019). A step-by-step guide to qualitative data coding. Routledge.

Albrecht, W. S., Clark, D. C., Smith, J. M., Stocks, K. D., & Woodfield, L. W. (1994). An accounting curriculum for the next century. *Issues in Accounting Education*, *9*(2), 401-425.

Albring, S. M., & Elder, R. J. (2020). Research initiatives in accounting education: Managing academic programs. *Issues in Accounting Education*, *35*(4), 61-74.

Alharahsheh, H. H., & Pius, A. (2020). A review of key paradigms: Positivism VS interpretivism. *Global Academic Journal of Humanities and Social Sciences*, *2*(3), 39-43.

Al-Htaybat, K., von Alberti-Alhtaybat, L., & Alhatabat, Z. (2018). Educating digital natives for the future: accounting educators' evaluation of the accounting curriculum. *Accounting Education*, *27*(4), 333-357.

Ali, I., Narayan, A. K., & Gedera, D. (2022). Transforming assessment in accounting education to align with online learning. *Pacific Accounting Review*, *34*(4), 536-547.

Arifin, S. R. M. (2018). Ethical considerations in qualitative study. *International journal of care scholars*, *1*(2), 30-33.

Arlinghaus, B. P. (2007). Accounting accreditation. The CPA Journal, 77(8), 69-72.

Babbie, E. R. (2020). The practice of social research. Cengage learning.

Baker, C. R., Karcher, J., & Tyson, T. (2007). Accounting advisory boards: A survey of current and best practices. <u>In:</u> *Advances in Accounting Education Teaching and Curriculum Innovations* (pp. 77-92). Emerald Group Publishing Limited.

Bampton, R., Cowton, C., & Downs, Y. (2013). The e-interview in qualitative research. In: Advancing research methods with new technologies (pp. 329-343). IGI Global.

Barrett, D., & Twycross, A. (2018). Data collection in qualitative research. *Evidence-based nursing*, *21*(3), 63-64.

Basit, T. (2003). Manual or electronic? The role of coding in qualitative data analysis. *Educational research*, *45*(2), 143-154.

Bauman, A. (2015). Qualitative online interviews: Strategies, design, and skills. *Qualitative research in organizations and management: An international journal*, *10*(2), 201-202.

Bazeley, P. (2009). Analysing qualitative data: More than 'identifying themes. *Malaysian journal of qualitative research*, *2*(2), 6-22.

Beetham, H., & Sharpe, R. (Eds.). (2019). *Rethinking pedagogy for a digital age: Principles and practices of design*. Routledge.

Bengio, Y., Louradour, J., Collobert, R., & Weston, J. (2009). Curriculum learning. <u>In:</u> *Proceedings* of the 26th annual international conference on machine learning (pp. 41-48).

Benigni, V., Ferguson, D., & McGee, B. (2011). Establishing a "renown-gown" relationship: The role of advisory boards in communication programs. *Journalism & Mass Communication Educator*, *66*(1), 54-68.

Berikol, B. Z., & Killi, M. (2021). The effects of the digital transformation process on the accounting profession and accounting education. <u>In</u>: *Ethics and Sustainability in Accounting and Finance, Volume II* (pp. 219-231). Springer, Singapore.

Beverland, M., & Lindgreen, A. (2010). What makes a good case study? A positivist review of qualitative case research published in Industrial Marketing Management, 1971–2006. *Industrial Marketing Management*, *39*(1), 56-63.

Bevir, M., & Hall, I. (2020). The English school and the classical approach: Between modernism and interpretivism. *Journal of International Political Theory*, *16*(2), 153-170.

Bingham, A. J., & Witkowsky, P. (2021). Deductive and inductive approaches to qualitative data analysis. *Analyzing and interpreting qualitative data: After the interview*, 133-146.

Boadi, I., & Osarfo, D. (2019). Diversity and return: the impact of diversity of board members' education on performance. *Corporate Governance: The International Journal of Business in Society*, *9*(4), 824–842.

Bonk, C. J., & Smith, G. S. (1998). Alternative instructional strategies for creative and critical thinking in the accounting curriculum. *Journal of accounting education*, *16*(2), 261-293.

Bowen, G.A. (2009). Document analysis as a qualitative research method. *Qualitative Research Journal*, 9(2), 27–40.

Buschle, C., Reiter, H., & Bethmann, A. (2021). The qualitative pretest interview for questionnaire development: outline of programme and practice. *Quality & Quantity*, 1-20.

Busher, H., & James, N. (2009). Online interviewing. Online Interviewing, 1-176.

Campbell, D., & Cowton, C. J. (2015). Method issues in business ethics research: Finding credible answers to questions that matter. *Business ethics: A European review*, *24*, S3-S10.

Cardano, M. (2020). *Defending qualitative research: design, analysis, and textualization*. Routledge.

Clark, V. L. P. (2019). Meaningful integration within mixed methods studies: Identifying why, what, when, and how. *Contemporary Educational Psychology*, *57*, 106-111.

Clay, M. E. (2013). Incorporating International Financial Reporting Standards: IFRS <u>In</u>: The Accounting Curriculum. *Proceedings of the Northeast Business & Economics Association*.

Cohen, J. R., Dalton, D. W., Holder-Webb, L. L., & McMillan, J. J. (2020). An analysis of glass ceiling perceptions in the accounting profession. *Journal of Business Ethics*, *164*(1), 17-38.

Connelly, L. M. (2016). Trustworthiness in qualitative research. *Medsurg Nursing*, 25(6), 435.

Conroy, P. A., Lefever, M. M., & Withiam, G. (1996). The value of college advisory boards. *Cornell hotel and restaurant administration quarterly*, *37*(4), 85-89.

Correa Ruiz, C. (2013). A commentary on 'integrated reporting: a review of developments and their implications for the accounting curriculum'. *Accounting Education*, *22*(4), 360-362.

Cotton, F. (2015). A journey into the lived experience: A review of Janet Salmon's qualitative online interviews: Strategies, designs, and skills. *The Qualitative Report*, *20*(1), 39-41.

Courtney, W. T., Hartley, B. K., Rosswurm, M., LeBlanc, L. A., & Lund, C. J. (2021). Establishing and Leveraging the Expertise of Advisory Boards. *Behavior Analysis in Practice*, *14*(1), 253-263.

Creswell, J. W. (2007). Qualitative inquiry and research design: Choosing among five approaches (3rd ed.). Thousand Oaks, CA: Sage Publications.

Creswell, J. W. (2021). A concise introduction to mixed methods research. Sage Publications.

Creswell, J. W., & Creswell, J. D. (2017). *Research design: Qualitative, quantitative, and mixed methods approaches.* Sage publications.

Creswell, J. W., & Plano Clark, V. L. (2011). Choosing a Mixed Methods Design. <u>In</u>: *Designing and Conducting Mixed Methods Research*. Thousand Oaks: Sage Publications.

Creswell, J. W., & Poth, C. N. (2016). *Qualitative inquiry and research design: Choosing among five approaches.* Sage publications.

Creswell, J.W. (2002) Educational Research: Planning, Conducting and Evaluating Quantitative and Qualitative Research. Upper Saddle River, NJ: Pearson Education.

Curry, D. S. (2020). Interpretivism and norms. *Philosophical studies*, 177, 905-930.

Dehghani, M., & Pakmehr, H. (2011). Managerial challenges of curriculum implementation in higher education. *Procedia-Social and Behavioral Sciences*, *15*, 2003-2006.

Dewua, K., & Barghathť, Y. (2019). The accounting curriculum and the emergence of Big Data. *Accounting and Management Information Systems*, *18*(3), 417-442.

Doyle, W. (1992). Curriculum & pedagogy. In Jackson, P. W. (Ed.), *Handbook of research on curriculum* (pp. 486–516). New York, NY: Macmillan.

Drennan, J. (2003). Cognitive interviewing: verbal data in the design and pretesting of questionnaires. *Journal of advanced nursing*, *42*(1), 57-63.

Dul, J., & Hak, T. (2007). Case study methodology in business research. Routledge.

Edeigba, J. (2022). Employers' expectations of accounting skills from vocational education providers: The expectation gap between employers and ITPs. *The International Journal of Management Education*, *20*(3), 100674.

Elliott, V. (2018). Thinking about the coding process in qualitative data analysis. *The Qualitative Report,* 23(11), 2850-2861.

Enache, L., & García-Meca, E. (2019). Board composition and accounting conservatism: The role of business experts, support specialists and community influentials. *Australian Accounting Review*, *29*(1), 252-265.

Erstad, O., & Voogt, J. (2018). The twenty-first century curriculum: issues and challenges. *Springer International Handbooks of Education*, 19-36.

Feld, M. M. (2002). Advisory Boards and Higher Education's Urban Mission: Four Models. *Metropolitan Universities*, *13*(1), 22-32.

Flick, U. (2018). Triangulation in data collection. *The SAGE handbook of qualitative data collection*, 527-544.

Flyvbjerg, B. (2006). Five misunderstandings about case-study research. *Qualitative inquiry*, *12*(2), 219-245.

Forster, M. (2004). Higher order thinking skills. *Research Developments*, 11(11), 10-15.

Fritz, R. L., & Vandermause, R. (2018). Data collection via in-depth email interviewing: Lessons from the field. *Qualitative health research*, *28*(10), 1640-1649.

Gabbin, A. L. (2002). The crisis in accounting education. Journal of Accountancy, 193(4), 81-86.

Gabbin, A. L., Irving, J. H., & Shifflett, E. M. (2020). Accounting Education at a Crossroads. *The CPA Journal*, *90*(9), 50-54.

Genheimer, D. S. R., & Shehab, D. R. L. (2009). A survey of industry advisory board operation and effectiveness in engineering education. *Journal of Engineering Education*, *98*(2), 169-180.

Gerring, J. (2004). What is a case study and what is it good for? *American political science review*, *98*(2), 341-354.

Gill, P., Stewart, K., Treasure, E., & Chadwick, B. (2008). Methods of data collection in qualitative research: interviews and focus groups. *British dental journal*, *204*(6), 291-295.

Gill, S. L. (2020). Qualitative sampling methods. Journal of Human Lactation, 36(4), 579-581.

Graves, K. (2021). Mind the gap: A tale of two curriculum fallacies. Language Teaching, 1-13.

Griffin, W. C. (2022). Perceptions of hospitality industry professionals and hospitality faculty on industry-academia collaborations. *Journal of Hospitality & Tourism Education*, *34*(2), 99-110.

Guba, E.G. & Lincoln, Y.S. (1994). Competing paradigms in qualitative research. *Handbook of Qualitative Research*, 2(105), 163-194.

Guest, G., Namey, E. E., & Mitchell, M. L. (2013). *Collecting qualitative data: A field manual for applied research*. Sage Publications.

Guimón, J. (2013). Promoting university-industry collaboration in developing countries. *World Bank*, *3*, 12-48.

Harms, T. (1974). Change-Agents in Curriculum. Young Children, 280-288.

Harrison, R. L., Reilly, T. M., & Creswell, J. W. (2020). Methodological rigor in mixed methods: An application in management studies. *Journal of Mixed Methods Research*, *14*(4), 473-495.

Hawkins, J. E. (2018). The practical utility and suitability of email interviews in qualitative research. *The Qualitative Report*, 23(2).

Healy, M., Hammer, S., & McIlveen, P. (2020). Mapping graduate employability and career development in higher education research: A citation network analysis. *Studies in Higher Education*, 1-13.

Heang, L. T., Ching, L. C., Mee, L. Y., & Huei, C. T. (2019). University education and employment challenges: An evaluation of fresh accounting graduates in Malaysia. *International Journal of Academic Research in Business and Social Sciences*, *9*(9), 1061-1076.

Heng, K., & Sol, K. (2023). Skills mismatches in Cambodian higher education: Major causes and possible solutions. *Cambodian Journal of Educational Research*, *3*(1), 76-91.

Hennink, M. M., & Kaiser, B. N. (2020). *Saturation in qualitative research*. Sage Publications Limited.

Hepperlen, R. A., Rabaey, P., Ngulube, E., Biggs, J., DeGracia, D., & Hearst, M. O. (2022). Aligning a training curriculum with local needs: Use of an advisory board. *International Social Work*, *65*(6), 1264-1273.

Ho, C. (2023). Fink's taxonomy applied to work-integrated learning: An'audit'of success strategies accounting students employ during recruitment. *International Journal of Work-Integrated Learning*, *24*(1), 1.

Hunt, I., Taylor, R., & Oberman, W. (2017). Advisory board engagement: Assisting undergraduates with resume development. *Journal of Education for Business*, *92*(6), 288-295.

Hwang, S. (2008). Utilizing qualitative data analysis software: A review of Atlas. ti. *Social Science Computer Review*, *26*(4), 519-527.

Imperial, M. T. (2005). Using collaboration as a governance strategy: Lessons from six watershed management programs. *Administration & Society*, *37*(3), 281-320.

Jackson, D., Michelson, G., & Munir, R. (2023). Developing accountants for the future: New technology, skills, and the role of stakeholders. *Accounting Education*, *32*(2), 150-177.

Jackson, D., Richardson, J., Michelson, G., & Munir, R. (2022). Attracting accounting and finance graduate talent–beyond the Big Four. *Accounting & Finance*, *62*(3), 3761-3790.

Jakubik, M. (2021). How can practical wisdom manifest itself in five fundamental management and leadership practices? *Vilakshan-XIMB Journal of Management*, *18*(1), 3-25.

Janghorban, R., Roudsari, R. L., & Taghipour, A. (2014). Skype interviewing: The new generation of online synchronous interview in qualitative research. *International journal of qualitative studies on health and well-being*, *9*(1), 24152.

Kaupins, G., & Coco, M. (2002). Administrator perceptions of business school advisory boards. *Education*, *123*(2), 351-357.

Kavanagh, M. H., & Drennan, L. (2008). What skills and attributes does an accounting graduate need? Evidence from student perceptions and employer expectations. *Accounting & Finance*, *48*(2), 279-300.

Kearns, K. (2019). The Leadership Portfolio Program at The University of Pittsburgh: Teaching leadership to graduate students1. *Teaching Public Administration*, *37*(3), 255-273.

Kekeya, J. (2016). Analysing qualitative data using an iterative process. *Contemporary PNG Studies*, *24*, 86-94.

Kennedy, K. M. (2019). Promoting the qualitative research approach in the discipline of forensic and legal medicine: Why more qualitative work should be promoted and how that can be achieved. *Journal of forensic and legal medicine*, *62*, 72-76.

Kent, J., Williamson, E., Goodenough, T., & Ashcroft, R. (2002). Social science gets the ethics treatment: research governance and ethical review. *Sociological Research Online*, *7*(4), 1-15.

Khan, S. N. (2014). Qualitative research method: Grounded theory. *International journal of business and management*, *9*(11), 224-233.

Kilcrease, K. M. (2011). Faculty perceptions of business advisory boards: The challenge for effective communication. *Journal of Education for Business*, *86*(2), 78-83.

Kivunja, C., & Kuyini, A. B. (2017). Understanding and applying research paradigms in educational contexts. *International Journal of higher education*, *6*(5), 26-41.

Kolev, K. D., Wangrow, D. B., Barker III, V. L., & Schepker, D. J. (2019). Board committees in corporate governance: a cross-disciplinary review and agenda for the future. *Journal of Management Studies*, *56*(6), 1138-1193.

Korstjens, I., & Moser, A. (2018). Series: Practical guidance to qualitative research. Part 4: Trustworthiness and publishing. *European Journal of General Practice*, *24*(1), 120-124.

Kosinski, M., Matz, S. C., Gosling, S. D., Popov, V., & Stillwell, D. (2015). Facebook as a research tool for the social sciences: Opportunities, challenges, ethical considerations, and practical guidelines. *American psychologist*, *70*(6), 543.

Kuckartz, U., & Rädiker, S. (2023). *Qualitative Content Analysis: Methods, Practice and Software*. Sage Publcations.

Lemon, L. L., & Hayes, J. (2020). Enhancing trustworthiness of qualitative findings: Using Leximancer for qualitative data analysis triangulation. *The Qualitative Report*, *25*(3), 604-614.

Lester, J. N., Cho, Y., & Lochmiller, C. R. (2020). Learning to do qualitative data analysis: A starting point. *Human resource development review*, *19*(1), 94-106.

Lincoln, Y., & Guba, E. G. (1985). Naturalistic inquiry. Newbury Park, CA: Sage Publications.

Linneberg, M. S., & Korsgaard, S. (2019). Coding qualitative data: A synthesis guiding the novice. *Qualitative research journal*, *19*(3), 259-270.

Lochmiller, C. R. (2021). Conducting thematic analysis with qualitative data. *The Qualitative Report*, *26*(6), 2029-2044.

Maali, B., & Al-Attar, A. M. (2020). Accounting curricula in universities and market needs: The Jordanian case. *SAGE Open*, *10*(1), 2158244019899463.

Mayer, I. (2015). Qualitative research with a focus on qualitative data analysis. *International Journal of Sales, Retailing & Marketing, 4*(9), 53-67.

McPeak, D., Pincus, K. V., & Sundem, G. L. (2012). The international accounting education standards board: influencing global accounting education. *Issues in Accounting Education*, *27*(3), 743-750.

Mello, J. A. (2019). Creating and developing effective business and professional school advisory boards. *Journal of Higher Education Theory and Practice*, *19*(2), 88-97.

57

Milman, A. (2017). The role and responsibilities of industry advisory boards in enhancing the educational experience. <u>In</u>: *Handbook of Teaching and Learning in Tourism* (pp. 549-563). Edward Elgar Publishing.

Mousavi, M., Zimon, G., Salehi, M., & Stępnicka, N. (2022). The effect of corporate governance structure on fraud and money laundering. *Risks*, *10*(9), 176.

Nicotera, A. M. (2019). Paradigms: Ways of knowing in organizational communication. In *Origins and Traditions of Organizational Communication* (pp. 71-86). Routledge.

Norman, C. S., & Bagranoff, N. A. (2019). Accounting advisory councils: Engagement between practice and academe. <u>In:</u> *Advances in Accounting Education: Teaching and Curriculum Innovations* (Vol. 23, pp. 133-151). Emerald Publishing Limited.

Nowell, L.S., Norris, J.M., White, D.E. & Moules, N.J. (2017). Thematic analysis: Striving to meet the trustworthiness criteria. *International Journal of Qualitative Methods, 16*(1), 1-13. 1609406917733847.

O'Connor, K. (2022). Constructivism, curriculum and the knowledge question: tensions and challenges for higher education. *Studies in Higher Education*, *47*(2), 412-422.

Olivero, N., & Lunt, P. (2004). When the ethic is functional to the method: the case of e-mail qualitative interviews. In *Readings in virtual research ethics: issues and controversies* (pp. 101-113). IGI Global.

Patton, M. Q. (2007). Sampling, qualitative (purposeful). *The Blackwell encyclopedia of sociology*. DOI:10.1002/9781405165518.wbeoss012.

Patton, M.Q. (2002). Qualitative research and evaluation methods. 3rd ed. Thousand Oaks, CA: Sage Publications.

Pope, C., & Mays, N. (2020). The role of theory in qualitative research. *Qualitative research in health care*, 15-26.

Priya, A. (2021). Case study methodology of qualitative research: Key attributes and navigating the conundrums in its application. *Sociological Bulletin*, *70*(1), 94-110.

Query, J. T. (2018). Actuarial science advisory boards: A survey of current and best practices. *Journal of Education for Business*, *93*(8), 403-411.

Rebele, J. E., & Pierre, E. K. S. (2019). A commentary on learning objectives for accounting education programs: The importance of soft skills and technical knowledge. *Journal of Accounting Education*, *48*, 71-79.

Reid, W. (2013). *Thinking About the Curriculum (Routledge Revivals): The nature and treatment of curriculum problems*. Routledge.

Richards, R., Stevens, R., Silver, L., & Metts, S. (2019). Overcoming employer perceptions of online accounting education with knowledge. *Administrative Issues Journal: Connecting Education, Practice, and Research, 8*(2), 70-80.

Rowley, J. (2012). Conducting research interviews. *Management research review*, 35(3/4), 260-271.

Ruona, W. E. (2005). Analyzing qualitative data. *Research in organizations: Foundations and methods of inquiry*, *223*(263), 233-263.

Ryan, G. (2018). Introduction to positivism, interpretivism and critical theory. *Nurse researcher*, *25*(4), 41-49.

Ryan, G. W., & Bernard, H. R. (2003). Techniques to identify themes. *Field methods*, *15*(1), 85-109.

Saldana, J. (2021). The coding manual for qualitative researchers. *The Coding Manual for Qualitative Researchers*, 1-440.

Saldaña, J. (2021). The coding manual for qualitative researchers. Sage Publications.

59

Samanthi, D., & Gooneratne, T. (2023). Bean counter to value-adding business partner: the changing role of the accountant and situated rationality in a multinational firm. *Journal of Accounting & Organizational Change*, *19*(3), 513-535.

Sant, E. (2019). Democratic education: A theoretical review (2006–2017). *Review of Educational Research*, *89*(5), 655-696.

Schwartz, B. N., & Fogg, S. L. (1985). Department of accounting advisory board: A method of communicating with the business and professional community. *Journal of Accounting Education*, *3(1)*, 179-184.

Senaratne, S., Gunarathne, N., Herath, R., Samudrage, D., & Cooray, T. (2022). Institutional pressures and responses to the introduction of integrated reporting into accounting curricula: the case of Sri Lankan universities. *Accounting Education*, *31*(5), 536-566.

Sieber, J. E., & Tolich, M. B. (2012). *Planning ethically responsible research* (Vol. 31). Sage Publications.

Smith, B. (2018). Generalizability in qualitative research: Misunderstandings, opportunities and recommendations for the sport and exercise sciences. *Qualitative research in sport, exercise and health*, *10*(1), 137-149.

Snead, K., Coleman, F., & McKinney, E. (2019). On the use of advisory board input for revising master of accountancy curriculum. <u>In</u>: *Advances in Accounting Education: Teaching and Curriculum Innovations* (Vol. 23, pp. 153-176). Emerald Publishing Limited.

Stake, R.E. (1995). The art of case study research. Thousand Oaks, CA: Sage Publications.

Stake, R.E. (2006.) Multiple case study analysis. New York: Guilford Press.

Stensaker, B., Jungblut, J., & Mihut, G. (2021). Strategic advisory boards-the emergence of shadow governance in universities? *International Journal of Leadership in Education*, 1-17.

Suskie, L. (2018). Assessing student learning: A common sense guide. John Wiley & Sons.

Tapis, G. P., & Delaney, J. S. (2017). Using accounting advisory boards: A way to promote collaboration between faculty and practitioners. *The CPA Journal*, *87*(9), 40-42.

Thanh, N. C., & Thanh, T. T. (2015). The interconnection between interpretivist paradigm and qualitative methods in education. *American journal of educational science*, *1*(2), 24-27.

Trung, N. K. Q. (2016). Professional ethics in accounting curriculum at a Vietnamese university. *International Journal of Management, Accounting and Economics, 3*(4), 261-271.

TUT. 2008. Policy on Academic Advisory Committees. Online available from: <u>https://jupiter.tut.ac.za/staffportal/index.php</u>. Accessed on: 3 September 2022.

TUT.2017.TUTTransformationFramework2017.https://www.tut.ac.za/Documents/2018/Transformation-Framework.pdfAccessedon:19October 2022.

TUT. 2022. Policy on Academic Advisory Committees. Online available from: <u>https://jupiter.tut.ac.za/staffportal/index.php</u>. Accessed on: 13 March 2023.

Twyford, E., & Dean, B. A. (2023). Inviting students to talk the talk: Developing employability skills in accounting education through industry-led experiences. *Accounting Education*, 1-23.

Van Hoof, H., Torres, A. G., Rodas, L. M. S., Eljurl, M. I., & Granda, M. J. (2017). Bridging the gap between industry and academia: Establishing an industry advisory board in Ecuador's higher education system. *Maskana*, *8*(1), 33-42.

Van Noy, M., Trimble, M. J., Jenkins, P. D., Barnett, E., & Wachen, J. (2012). Structure in community college career-technical programs: A qualitative analysis.

Vasileiou, K., Barnett, J., Thorpe, S., & Young, T. (2018). Characterising and justifying sample size sufficiency in interview-based studies: systematic analysis of qualitative health research over a 15-year period. *BMC medical research methodology*, *18*(1), 1-18.

Wickham, H., & Wickham, H. (2016). *Data analysis* (pp. 189-201). Springer International Publishing.

Wicks, D. (2017). The coding manual for qualitative researchers. *Qualitative research in organizations and management: an international journal, 12*(2), 169-170.

Willcocks, M. A. (2017). Curriculum as a Vehicle for Agency in Gifted Learners. *Kairaranga, 18*(1), 47-52.

Williams, E. N., & Morrow, S. L. (2009). Achieving trustworthiness in qualitative research: A panparadigmatic perspective. *Psychotherapy research*, *19*(4-5), 576-582.

Wrigley, T. (2018). 'Knowledge', curriculum and social justice. *The Curriculum Journal*, *29*(1), 4-24.

Xu, H., Waldrup, B. E., & Michenzi, A. (2023). Assessing the effectiveness of advisory boards in accounting programs. *Journal of Accounting Education*, *63*, 100847.

Yastrebov, O. A. (2016). Renewing University-Based Curriculum in Line with Societal Needs: A Case of Legal Education in Russia. *International Journal of Environmental and Science Education*, *11*(16), 9010-9016.

Yee, C. (2020). Can we be partners? Industry/corporate advisory boards for academic technical communication programs. <u>In:</u> *Publications management: Essays for professional communicators* (pp. 203-211). Routledge.

Yin, R. K. (2011). Applications of case study research. Sage Publications.

Yin, R. K. (2018). Case study research and applications. Sage Publications.

Yin, R.K. (2009). Case study research: Design and methods. 4th ed. Los Angeles: Sage Publications.

Zahra, S. A., Newey, L. R., & Shaver, J. M. (2011). Academic advisory boards' contributions to education and learning: Lessons from entrepreneurship centers. *Academy of Management Learning & Education*, *10*(1), 113-129.

Zimmerman, A., Brooker, N., & Mokma, A. (2000). Demographic and Personality Type Characteristics of Advisory Committee Members at a Techinical College. *NACTA Journal*, 45-55.

ANNEXURE 1



Thesis title

EXPLORING FACULTY PERCEPTIONS ON THE VALUE OF ADVISORY BOARDS IN ACCOUNTING EDUCATION CURRICULUM

KGOBALALE NEBBEL MOTUBATSE

Master of Business Administration (MBA) Education Management Master's thesis 2024



EXPLORING FACULTY PERCEPTIONS ON THE VALUE OF ADVISORY BOARDS IN ACCOUNTING EDUCATION CURRICULUM

by

NEBBEL MOTUBATSE

The interviews are consisting of questions, uniquely developed for the sole purpose of the study.

Interview Guide

The interviews consist of questions, uniquely developed for the sole purpose of the study.

Section A

1) Demographic information

Position of the participant	
How long in the position	
Position previously held	
How long was the previous position held?	

2) Descriptive information

Please indicate your highest level of academic qualification:

Grade12/matric	
Tertiary certificate/Diploma	
Degree	
Postgraduate degree	

Section B

Primary research questions

The research questions will be organized to explore three main areas:

Role of advisory boards

• How do participants perceive the role of advisory boards in the accounting curriculum?

Composition of advisory boards

• What is the composition of advisory boards in the faculty?

Challenges of the curriculum in the faulty

• What are the challenges of the curriculum in the faculty?

There are seven supporting questions:

1. Describe your views of the role of advisory boards in the accounting curriculum.

Probing question:

- What does this mean for the faculty? And to the university?
- 2. How is the composition of advisory boards in the faculty?

Probing question:

- How do you recruit the advisory boards?
- What qualities are you looking out for?
- 3. Do you have alumni/graduates on your current advisory board?
 - Do you think this could be improved?
 - How do you think you could improve this?
- 4. Have you ever felt that the advisory boards are helping you in improving the qualification programme and curriculum offering? What supports are they providing to improve curriculum offerings?

Probing question:

- Do you find these supports are ...?
 - Helpful?
 - Implementable?
 - Effective?
- Do you think faculty conditions allow advisory boards to exercise their role?
- How else do you think you could improve the current conditions?
- 5. Do you think advisory boards support our students enough to get industry exposure and employment?
- **6.** What challenges do you think prevent advisory boards from supporting the faculty programme?

Probing question:

- How could such challenges be reduced or overcome?
- 7. Are there any other areas that are relevant to this topic that you have not shared and would like to address now?

Thank you for your participation!

NB: These questions are developed for the exclusive purposes of the study.